Author's response to reviews

Title: Unravelling the impact of ethnicity on health in Europe: the HELIUS study

Authors:

  Karien Stronks (k.stronks@amc.uva.nl)
  Marieke B Snijder (m.b.snijder@amc.uva.nl)
  Ron JG Peters (r.j.peters@amc.uva.nl)
  Maria Prins (mprins@ggd.amsterdam.nl)
  Aart H Schene (a.h.schene@amc.uva.nl)
  Aeilko H Zwinderman (a.h.zwinderman@amc.uva.nl)

Version: 2  Date: 28 February 2013

Author's response to reviews:

Amsterdam, February 28 2013

Dear Editor,

Thank you very much for your e-mail dated February 21, concerning our paper ‘Unravelling the impact of ethnicity on health in Europe: the HELIUS study’ (MS: 1898748276920793).

We have now changed the manuscript according to your comments:

1. ABSTRACT: We have now formatted our abstract (as well as the first page and the references) according to the guidelines. We apologize for the inconvenience.

2. We confirm that there are no other related manuscripts to this study under consideration or published elsewhere.

3. We have now included a power calculation within the manuscript (page 18).

We look forward to the outcome of the peer review process.

Yours sincerely, on behalf of the co-authors,

Karien Stronks
Dept. of Public Health
Academic Medical Center
University of Amsterdam
Amsterdam
The Netherlands